CENWP-OP-B                                                                                                                                     31 May 2004

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 22 of 2004, which covers the period 23 May through 29 May 2004.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 244.4 to 311.7 kcfs.  Daily project tailwater elevation averaged 18.3’ to 22.2’ msl.  Daily project forebay elevation averaged 75.3’ to 75.9’ msl.  Daily river temperature averaged 56 to 57° F.  Daily turbidity measured 3.5 to 5.0 on the secchi disk.  

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B Branch respectively, are part of the FPP spill pattern.  Spill averaged 99.1 to 101.7 kcfs.    
1.3 Unit operation.   

1.3.1 Unit 1 remains out of service for turbine rehab.

1.3.2 Unit 2 had a 1% violation from 0908 until 0928 on 24 May for pre-rehab testing.

1.3.3 Unit 6 went out of service from 0500 on 26 May until 1515 on 27 May for annual maintenance.  

1.3.4 Unit 12 had a 1% violation from 1007 until 1027 on 25 May due to STS inspections.

1.3.5 Unit 13 had a 1% violation from 1030 until 1059 on 25 May due to STS inspections.

1.3.6 Unit 15 went to manual 3D cam adjustment at 0933 on 05 April.  The cam was automatically adjusting and draining power which led to a governor lock up and complete loss of control.  New modules are on order.   

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  

1.5 Gatewell drawdowns are measured on Wednesdays.  Drawdowns are measured daily on Unit 17.

2.  ACTIVITIES. 
2.1 AWS Closures.  The following AWS valves were closed this reporting week.

FV1-1 - closed for cleaning from 1320 until 1330 on 25 May.

2.2 STS/VBS Inspections.  The following STS/VBSs were inspected this reporting week.

	Unit
	Date
	Run Hours
	Comments

	12

13

15

16

17
	5/25

5/25

5/25

5/20

5/24
	550

537

567

136

       344
	Will video ASAP

C-slot covered for Gap Loss test.  Will video when test is complete.

Washed VBS in 17-B


2.3 Dewatering and Fish Salvage.  No dewatering or salvage activities occurred this reporting week. 

3.  RESEARCH.
3.1 Battelle/BAE Systems FGE testing.  Data collection continues.     

3.2 Battelle/BAE Systems Gap Loss Testing.  Data collection continues through 3 June. 

3.3 CRITFC.  Sampling twice a week.

3.4 ERDC TDG.  Sensors remain installed in the tailrace.

3.5 ODFW Pikeminnow Tagging.  Electrofishing in Bonneville’s forebay and tailrace will continue on 28 June through 2 July.

3.6 NMFS.  640-clipped yearling chinook were collected for NMFS researchers during the week.

3.7 NOAA FISH lamprey.  Lamprey ladder installation began 24 May.  

3.8 PSMFC SMF.  Three steelhead slid over the separator bars this week.  One kelt was stuck on separator bars and released alive.   

3.9 USGS FPE testing.  Fish tagging and releasing continues.  PSMFC collected 635-clipped yearling chinook and 645-clipped steelhead for USGS RT testing.  

3.10 University of Idaho.  Sampling daily in the AFF.

3.11 Wildlife Services/Avian Abatement.  Birds were hazed this reporting week with rockets, cracker shells, and 15mm pyrotechnics.  None were lethally taken.

3.12 WDFW Fish Counting.  16-hr visual counts at Washington Shore and Bradford Island.

4.  FISHWAYS 
4.1 Project biologists inspected 23 May through 29 May this reporting week.  Larry Basham with FPC inspected 24 May.    

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR#

	All week
	PH1 PCC
	Velocity <1.5 fps
	Meter in slowest part of the channel.
	This is a normal condition.
	

	All week
	FG3-4
	Incorrect position for TW
	Bent shaft
	No action taken.
	

	All week
	PH2 PCC
	Velocity < 1.5 fps
	Unknown
	Visual inspection in CC next week.
	

	5/23-5/24
	SUE, SDE
	Tailwater-gate < 13’
	TW reading is wrong on SA24 board 
	Recalibration of TW sensor
	

	5/23-25 and 5/27
	Weir 37
	Weir reading >1.3’ 
	Shad 
	See Misc. 6.2 
	

	5/24
	Weir 38
	Weir reading >1.3’
	Shad 
	See Misc. 6.2
	

	All week
	FV 5-9
	In manual
	Stillwell plugged
	TR submitted

See 4.6.2
	04-19985

	5/29
	Floating orifice gate #7
	Stuck submerged
	Unknown
	No action taken 

See 4.6.3
	


4.3 Fish Ladder Temperatures.   Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Temperatures are in Fahrenheit.

	Date
	Wash. Shore
	Bradford Is.

	5/23
	57.9
	57.9

	5/24
	58.6
	58.7

	5/25
	58.9
	59.0

	5/26
	59.0
	59.2

	      5/27
	59.2
	59.3

	      5/28
	58.6
	58.8

	5/29
	        58.4
	58.4


4.4 Sensor Calibration.  Calibration was checked on 28 May this reporting week.

4.5  Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

4.6 Fishway Status.    

4.6.1 DSM2 orifices 12 C-S and 16 B-N were manually operated from 0700 to 1353 on 24 May for shaft/bracket repairs.

4.6.2 FV 5-9 has not been properly controlling in Auto during this week.  On 27 May, Operators and garage crew attempted to flush the stillwell using the fire truck.   TR has been submitted to the structural crew for cleaning.  

4.6.3 Floating orifice gate (FO-7) was stuck submerged with a trickle of water over the top on 29 May.  By 31 May, gate had returned to normal operating position.     

5 CONSTRUCTION

5.1 PH 2 Corner Collector Hi-Q PIT tag detector- Prototype antenna construction started on Cascades Island.  If feasible, the prototype antenna will then be constructed into the existing channel during the in-water work period during FY05.  See Misc. 6.3 

5.2 SMF hydrocannons.  The hydro cannons at the SMF are not operational at this time.  The Project is working on repairs.    

5.3 Adult PIT tag exit detectors.  Current schedule is set for installation in WA ladder for Winter 04/05.

5.4 Head Gate Delivery.  Off loading of headgates at PH1 is scheduled for 7 June 2004.

5.5 PH 1 Gantry Crane Delivery and Erection.  Scheduled for 9 August through 4 December.    

6 MISCELLANEOUS AND OTHER NOTEWORTHY ITEMS

6.1 Sea lions and harbor seals have been observed in the tailrace of both powerhouses and the spillway.  No action is being taken by the project to harass, harm, or otherwise annoy these animals.  

6.2 Shad appear to be clogging the bleed off valve for weir 37.  Operators to blow air through the screen twice daily and structural crew to rake on Monday’s and Thursday’s if needed. 

6.3  PIT antenna scheduled for delivery at PH 2 forebay on 24 June.  

6.4 On 22 May, a new Teletype revising unit priority was effective on 23 May to facilitate research testing for FGE and survival.  

6.5 On 24 May, a new Teletype, effective 26 May, indicated a forebay and tailwater restriction for hydraulic testing at Powerhouse 2 collection channel.   

6.6 On 28 May, a new Teletype indicating an increase to 120% TDG for the evening to 160 Kcfs. 

JAMES R. MAHAR, P.E.

Operations Manager
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